Author's response to reviews

Title: Symptomatic Cushing syndrome and hyperandrogenemia revealing steroid cell ovarian neoplasm with late intra-abdominal metastasis

Authors:

Menghua Yuan (13752729642@163.com)
Mingcai Qiu (tjmughen@163.com)
Mei Zhu (meichuqin@163.com)

Version: 3 Date: 6 February 2014

Author's response to reviews:

Dear editor and reviewers,

Thank you for your kind review. I have revised our manuscript in red fonts in light of the reviewers' comments.

For Reviewer: Bernard Conte-Devolx,
1. I have revised the title and made it include the fact that the tumor was secreting cortisol and androgens;
2. I have revised the discussion to emphasize the discrepancy between clinics and biology first, and the first result of pathology, in a way that a second pathology reading should have been performed immediately after the first surgery, which should have led to an appropriate and strict follow-up.

For Reviewer: Fahrettin Kelestimur,
1. I have corrected these spelling errors like “hirsuitism”, “His” and “µg/ml “;
2. I have corrected the wrong statement “… the patient had typical clinical pictures of Cushing’s syndrome” in the discussion;
3. I have discussed the issue whether routine screening for Cushing’s syndrome is required in women presented with hirsutism, referring to the recent paper (Karaca et al. EJE, 2013).

At last, I have done some language corrections to make our manuscript more fluently.

Best wishes,

Sincerely yours

Mingcai Qiu